Author's response to reviews

Title: Is different better? Models of teaching and their influence on the net financial outcome for general practice teaching posts

Authors:

  Caroline O Laurence (caroline.laurence@adelaide.edu.au)
  Linda E Black (linda.black@adelaide.edu.au)
  Carolyn Cheah (carolyn.cheah@adelaide.edu.au)
  Jonathan Karnon (jonathan.karnon@adelaide.edu.au)

Version: 3 Date: 9 June 2011

Author's response to reviews: see over
9 June 2011

Editor
BMC Medical Education

Dear Editor

Thank you for the opportunity to review our manuscript entitled *Is different better? Models of teaching and their influence on the net financial outcome for general practice teaching posts* (MS 1955476607526572).

As requested we have responded to the reviewers’ comments and the changes made are outlined below. The changes to the paper have been made using the “Track Changes” function.

If you have any other queries, please do not hesitate to contact me on.

With kind regards

Caroline Laurence
Research and Development Manager